Author's response to reviews

Title: Prevalence and risk factors of chronic kidney disease in urban adults Cameroonian according to three common estimators of the glomerular filtration rate: A cross-sectional study

Authors:

Francois F Kaze (francoiskaze@yahoo.fr)
Marie-Patrice Halle (patricehalle@yahoo.fr)
Hermine T Mopa (mopahermine@yahoo.fr)
Gloria Ashuntantang (maglo09@hotmail.com)
Hermine Fouda (mendjouf@yahoo.fr)
Jeanne Ngogang (jngogang@yahoo.fr)
Andre-Pascal Kengne (apkengne@yahoo.com)

Version: 4 Date: 30 May 2015

Author's response to reviews:

MS: 1432287839153531
Prevalence and risk factors of chronic kidney disease in urban adults Cameroonian according to three common estimators of the glomerular filtration rate: A cross-sectional study
Response to reviewer’s comments
Dear Editor,

With reference to our manuscript currently undergoing consideration for possible publication in the Journal, we welcome the opportunity of replacing the term renal failure in the text to avoid any confusion in the literature.

In line with the Editor’s advice, we have now used the term “decreased GFR” to represent any “eGFR<60 ml/min” and have maintained the term “CKD” to represent the presence of “any albuminuria and/or eGFR<60 ml/min”, throughout the revised manuscript. We have also completely updated the format of the manuscript to comply with the Journal’s style.

We look forward to the outcome of your assessment.
For the co-authors,
Dr Kaze